Author’s response to reviews

Title: A Biomechanical Analysis of Triangular Medial Knee Reconstruction

Authors:
xiaomeng wang (hbydwxm@163.com)
Huixin Liu (18631808787@163.com)
Guman Duan (40002208@qq.com)
Yingzhen Niu (missniu99@163.com)
Chang Liu (1158344848@qq.com)
Fei Wang (18633835079@yeah.net)

Version: 1 Date: 24 Oct 2017

Author’s response to reviews:

Dear Editors: Thank you for your letter and for your comments concerning our manuscript (BMSD-D-17-00978R1).

Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval.